Author’s response to reviews

Title: Antioxidant properties of different solvent extract from Carpobrotus edulis (L.) Bolus leaf used for the management of common infections in HIV/AIDS patients in Eastern Cape Province

Authors:

Beauty B E Omoruyi (sayhi2sa@yahoo.com)
Anthony A J Afolayan (aafolayan@ufh.ac.za)

Version: 9 Date: 14 August 2012

Author’s response to reviews: see over
Dear Editor,

Beauty et al.

Thank you very much for your patient and kind gesture. We have edited our manuscript titled “Antioxidant properties of different solvent extract from *Carpobrotus edulis* (L.) Bolus leaf used for the management of common infections in HIV/AIDS patients in Eastern Cape Province” Beauty et al. 1456165686768294 according to your Journal. The figures, tables and a cover letter are uploaded as separate files.

Best regards

Beauty.